**Author's response to reviews**

**Title:** TICOPA: Tight COntrol of Psoriatic Arthritis A randomised controlled trial to compare intensive management vs standard care in early psoriatic arthritis

**Authors:**

Laura Coates (l.c.coates@leeds.ac.uk)
Nuria Navarro-Coy (n.navarro-coy@leeds.ac.uk)
Sarah R Brown (s.brown@leeds.ac.uk)
Sarah Brown (medsbro@leeds.ac.uk)
Lucy McParland (l.mcparland@leeds.ac.uk)
Howard Collier (h.a.collier@leeds.ac.uk)
Emma Skinner (e.skinner@keele.ac.uk)
Jennifer Law (j.a.law@leeds.ac.uk)
Anna Moverley (A.R.Caperon@leeds.ac.uk)
Sue Pavitt (s.pavitt@leeds.ac.uk)
Claire Hulme (c.t.hulme@leeds.ac.uk)
Philip G Conaghan (p.conaghan@leeds.ac.uk)
Paul Emery (p.emery@leeds.ac.uk)
Philip Helliwell (p.helliwell@leeds.ac.uk)

**Version:** 2 **Date:** 20 February 2013

**Author's response to reviews:**

Dear Rey

Please see the points below in response to your request for updates to the manuscript submission:

1) The trial registration number is now corrected to 'ISRCTN30147736'

2) I can confirm that no data analysis has yet been conducted on the trial and we are not currently preparing other manuscripts based on this study.

3) Proofs of funding and ethics are not uploaded with the revised manuscript.

In addition, the references in the revised manuscript now conform to the journal style specified in the authors' checklist.

Kind Regards

Sarah